Author’s response to reviews

Title: Investigation of Associated Factors with Post-Operative Outcomes in Patients Undergoing Tetralogy of Fallot Correction

Authors:

M.H. Nemati (Nemati@gmail.com)

H. Badie (Badie@YAHOO.COM)

Mansour Jannati (mansour.jannati@mail.com)

Version: 1 Date: 05 Nov 2017

Author’s response to reviews:

The manuscript is well written. I suggest the authors to discuss in the manuscript these interesting papers.

PMID: 25245139, 24655794, 28935073
the first suggested reference is added.

Reviewer reports:

Giuseppe Filiberto Serraino (Reviewer 1): The authors investigate potential factors related with outcomes after TOF’s surgery.

I would suggest to expand the Result section and the paragraph about the statistical methods.

Result is expanded. Statistical analysis section is exist in the method section.

I think it would be interesting if the authors could identify a cut off value for the haematocrit level the can suggest a better or worse outcome.

A sentence is added about this comment.

Raffaele Serra (Reviewer 2): Good and timely article on an important issue. Overall the study is very well conducted and written. I accept it in the current format.

Thank you.

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing
service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

I tried to correct the language problems.